Author’s response to reviews

Title: Clinical and microbiological characteristics and outcomes of community-acquired sepsis among adults: a single center, 1-year retrospective observational cohort study from Hungary

Authors:

Balint Szabo (szabo.balint.gergely@gmail.com)
Rebeka Kiss (kefalika@gmail.com)
Katalin Lenart (katalenart@gmail.com)
Bence Marosi (m.bence10@gmail.com)
Eszter Vad (vadeszter@gmail.com)
Botond Lakatos (botondt.lakatos@gmail.com)
Eszter Ostorházi (ostorhazi.eszter@med.semmelweis-univ.hu)

Version: 2 Date: 13 Jun 2019

Author’s response to reviews:

Dear Editor,

We would like to again thank the Editorial Staff and the Reviewers for their fast work and suggestions, and resubmit the finalized version of the manuscript for publication in BMC Infectious Diseases.

We carried out minor revisions according to the recommendations given by the Editor below this letter. Reviewers #1 and #2 did not specify any points for revision which we could find.

Please find the revised manuscript attached.

Thank you for your kind reconsideration in advance.

Yours sincerely,
Eszter Ostorhazi, M.D. PhD (corresponding author)

Balint Gergely Szabo, M.D. (first author)

1. Please format your abstract under the following headings: Background, Methods, Results, Conclusions, Keywords. More information on what should be included in these sections can be found at https://bmcinfectedis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article – The abstract was formatted accordingly.

2. Please clarify whether any administrative permissions were required to access the raw data. If so please state who granted permission in the “Ethics approval and consent to participate” section. – The section was extended to include information on the administrative permissions.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. – All tracked changes were removed in MS Word.